Sodium Borohydride

A Study of the Effect of Operational Conditions on the Material Characteristics of Salt-like Substances During Bulk Storage

BEHLEN

ME54010-20: Literature Research Report number: 2023.MME.8834. Casper E. van Eijk

elft

of Technoloc

Sodium Borohydride

A Study of the Effect of Operational Conditions on the Material Characteristics of Salt-like Substances During Bulk Storage

⊳ Casper E. van Eijk

Student Name

Student Number

Van Eijk, Casper E. 4599756

Supervisor:Dr. Ir. D. L. SchottSupervisor:Ir. M. C. van BentenFaculty:Faculty of Mechanical Engineering, Delft

Cover: Photographer unknown (n.d.). Grayscale photo of silos, ragan, nebraska, grain elevator, rural, HD wallpaper (Modified) [Photograph]. https://www.wallpaperflare.com/grayscale-photo-of-silosragan-nebraska-grain-elevator-rural-wallpaper-zvrhg



Summary

Throughout this literature research report, the influence of operational conditions during bulk storage on the mechanical characteristics of various inorganic, crystalline, day-to-day salts has been investigated, in order to make a prediction on the effect of operational conditions on the mechanical characteristics of sodium borohydride. Firstly, an analysis and selection of the various mechanical characteristics has been conducted. Secondly, an analysis of the chemical properties helped to select a list of materials that are chemically similar to sodium borohydride. Thirdly, an effort was made to compare these materials to sodium borohydride using the selected mechanical characteristics. Using the particle sizes, bulk densities and angles of repose, sodium borohydride could to a certain extent be linked to calcium chloride, magnesium sulfate, potassium chloride, potassium carbonate and silicon dioxide. Lastly, an analysis on the effect of operational conditions on the mechanical characteristics of the materials has been conducted. No definitive conclusions could be drawn about the effect of the operational conditions on the mechanical characteristics of sodium borohydride.

Contents

Summary i											
1	Introduction 1.1 Purpose of Research 1.2 Research Questions 1.3 Methodology and Scope of Research 1.4 Outline of Research	1 1 2 2									
2	Mechanical Characteristics 2.1 Classes of Mechanical Characteristics 2.2 Definitions of Bulk Material Characteristics 2.3 Selection of Relevant Bulk Material Characteristics	4 4 5 9									
3	Resembling Materials3.1Analysis of the Chemical Properties of Sodium Borohydride3.2Comparable Inorganic, Crystalline, Salt-like Materials3.3Mechanical Characteristics of these Materials	10 10 11 12									
4	Sodium Borohydride4.1Comparison between Materials and Sodium Borohydride4.2Conclusions	13 13 16									
5	Operational Conditions 5.1 Definitions of the Operational Conditions 5.2 Influence of the Operational Conditions 5.2.1 Sodium Chloride 5.2.2 Calcium Carbonate 5.2.3 Sodium Bicarbonate 5.3 Conclusions	17 17 18 19 20 20 21									
6	Conclusion	22									
Re	References 2										
Α	Additional Information A.1 Search Strategy	33 33									

Introduction

Nowadays, almost everyone is aware of global warming and the implications that accompany this effect. It is the responsibility of humans to try to minimize this effect in any way possible. One of the main contributors to this global warming effect is the colossal amount of energy that is consumed every day. The consumption of fossil fuels especially, is detrimental to the environment [1]. Because of this, it is imperative to try to decrease the amount of energy that is consumed and to try to find new, greener ways to fulfil the remaining energy demand. Climate change targets have highlighted the importance of finding solutions to these challenges and one of the solutions to keeping up with the high demand of energy sustainably is to make use of hydrogen produced power, a renewable energy source that only emits water [2]. Hydrogen is a clean fuel, especially when the electrolysis of water to create the hydrogen is done with renewable electricity. On top of that, the technology that uses hydrogen fuel to work is often a lot more efficient. However, hydrogen gas is a difficult gas to handle. It needs to be highly compressed to be transported in gas form, or it needs to be kept at an extremely low temperature to be transported in liquid form [3, 4]. It becomes apparent that finding other ways of storing hydrogen are necessary.

1.1. Purpose of Research

The need for other ways of storing hydrogen, in order to provide enough renewable energy solutions to keep up with the energy demand, creates possibilities for new subjects to study. One of these subjects is the storage of hydrogen using sodium borohydride ($NaBH_4$) as the storage medium. According to Mohring and Wu [3], the hydrogen on demand system that has been developed and patented by Millennium Cell proves to be an intelligent way of using the sodium borohydride as an easily controllable storage medium for practical hydrogen generation possibly in applications for backup power or fuel cell vehicles. However, a significant amount of work still needs to be done in order to bring this technology to commercialization [3]. A particular area that still needs a significant amount of research is the behaviour of sodium borohydride during bulk storage, with emphasis on the behaviour of the material throughout a range of different operational conditions.

The purpose of this literature research is to investigate the behaviour of sodium borohydride during bulk storage while subjected to various operational conditions. Due to the limited data available on the behaviour of the bulk stored sodium borohydride, this literature research tries to accomplish this by comparing sodium borohydride to various other materials that bear a close resemblance to sodium borohydride and possess relatively similar mechanical characteristics.

1.2. Research Questions

In order to successfully accomplish the purpose of this research, the following research question will have to be answered:

What is the effect of different operational conditions during bulk storage on the mechanical characteristics of materials similar to sodium borohydride? This research question has been divided into the following sub-questions that will outline this report:

- 1. Which mechanical characteristics are of importance in classifying bulk materials?
- 2. Which materials bear a resemblance to sodium borohydride?
- 3. To what extent can these materials be compared to sodium borohydride?
- 4. What is the effect of bulk storage operational conditions on the mechanical characteristics?
- 5. What is the expected behaviour of sodium borohydride to the bulk storage operational conditions?

1.3. Methodology and Scope of Research

In order to answer these sub-questions, and subsequently answer the research question, a certain stepby-step methodology has been followed. Firstly, a selection of the relevant mechanical characteristics has been made. With respect to these mechanical characteristics, the aim is on the bulk material characteristics, describing the mechanical characteristics of the bulk material in interaction with itself. The particle material characteristics, describing the material characteristics of one particle, have been omitted because this research deals with bulk material, rather than individual particles. The equipment material characteristics and the interaction characteristics have been omitted because in this research, the presence of handling equipment has been neglected.

Secondly, a list of twenty materials bearing a resemblance to sodium borohydride has been constructed. For the selection of these materials, the focus has been on inorganic, crystalline, salt-like materials. Additionally, it has been decided that these materials need to have a day-to-day application. This is decided because it is expected that a lot of information is available on the production and storage of these materials. After the selection of the relevant materials, selected mechanical characteristics have been listed.

Thirdly, a comparison has been made between sodium borohydride and the other materials. And lastly, an analysis has been conducted on the effect of the operational conditions on the mechanical characteristics of the twenty materials. Information has only been found on this effect for three of the twenty materials, so to illustrate the information gap present in the available literature, a detailed search strategy has been included in Appendix A.

1.4. Outline of Research

This literature research is guided by the aforementioned sub-questions. As a result, the following chapters have been created.

- Chapter one includes the purpose and research questions, as well as the methodology, scope, and outline of this research report.
- Chapter two includes definitions of the various classes of mechanical characteristics and a selection of the relevant classes, as well as definitions of the relevant mechanical characteristics in these classes.
- Chapter three includes an introduction to sodium borohydride, as well as the selection process
 of the twenty materials and a list of the mechanical characteristics of these materials.
- Chapter four includes a comparison between sodium borohydride and the twenty materials based on their mechanical characteristics.
- Chapter five includes an analysis of the operational conditions that become relevant when storing bulk material. It includes definitions of the operational conditions, as well as the influence of these operational conditions on the relevant mechanical characteristics of the resembling materials. On top of that, a prediction is made on the influence of the operational conditions on the mechanical characteristics of sodium borohydride.
- In chapter six, the conclusions of this literature research have been summarized, and the research questions have been answered.



Figure 1.1: Outline of this report.

 \sum

Mechanical Characteristics

In this chapter, an introduction is given into the mechanical characteristics that are widely used to describe and define materials and their behaviour. First, the different groups of mechanical characteristics are introduced and defined, and the relevant groups are selected for further analysis. Next, the mechanical characteristics in the selected groups are defined, and the relevant characteristics are determined.

2.1. Classes of Mechanical Characteristics

Mechanical characteristics describe the characteristics and behaviour of certain materials and their surroundings. The mechanical characteristics can be divided into four different classes. The final summary of these classes and its characteristics can be found in Figure 2.1 [5][6][7].

Mechanical characteristics

Class 1: *Particle material characteristics,* describing the material characteristics of one particle. **Class 2:** *Bulk material characteristics,* describing the material characteristics of a bulk of material interacting with itself.

Class 3: *Equipment material characteristics,* describing the material characteristics of the handling equipment that is in contact with the material.

Class 4: *Interaction characteristics,* describing the characteristics of the interaction between handling equipment and bulk material.

Class 1: Particle Material Characteristics

The first and most straightforward class of mechanical characteristics is the class of particle material characteristics. Particle material characteristics describe the material characteristics of a single particle. Common particle material characteristics are particle shape and size, density, elasticity, and Poisson's ratio, internal damping, surface roughness and surface hardness.

Class 2: Bulk Material Characteristics

Bulk material characteristics describe the mechanical characteristics and the material behaviour of a bulk of multiple particles interacting with one another. Common bulk material characteristics are the particle size and shape distributions, the bulk density and elasticity, the compressive and tensile strength, the internal friction, the shear angle, the angle of repose, cohesion and flowability of the bulk.

Class 3: Equipment Material Characteristics

The third class consists of the mechanical characteristics of the handling equipment itself. Common equipment material characteristics are the shape and size of the equipment, the displacements, velocities, and accelerations of the equipment, the surface roughness of the equipment and mechanical properties such as the shear modulus, Poisson's ratio and the hardness of the equipment material.



Figure 2.1: Overview of the mechanical characteristics.

Class 4: Interaction Characteristics

The interaction characteristics describe the mechanical characteristics of the interaction between the handling equipment and the material itself. Common interaction characteristics are the coefficients of static and dynamic friction between the particles and the equipment, the adhesion between particles and equipment and the characteristic size of the equipment with respect to the particle size.

Selection of Relevant Class of Mechanical Characteristics

Not all of these four classes of mechanical characteristics are of particular interest to this literature assignment. Class 1 has been omitted from this research because it concerns itself with the material characteristics of individual particles, rather than those of bulk material. Class 3 and 4 have been omitted due to the fact that this research does not include the influence of the equipment on the bulk material. Class 2 on the other hand is of great interest to this research, because this class concerns the bulk material characteristics.

2.2. Definitions of Bulk Material Characteristics

As was discussed before, the bulk material characteristics are the most important mechanical characteristics to this literature research. This is due to their applicability to bulk material, and its behaviour during bulk storage. Within the class of bulk material characteristics, a distinction can be made between eleven individual characteristics. An introduction and definition to these eleven characteristics has been provided.

Particle Size Distribution

A lot of different types of granular materials exist in an array of different applications and varieties. Bulks of these granular materials almost always consist of groups of particles with different particle sizes existing within this bulk in a sort of distribution, also called the particle size distribution [8]. When using the particle size or particle size distribution, it is important to take note of the way in which this size is determined. Vastly different ways of selecting a relevant measurement length can be distinguished throughout the literature [9, 10].

When the size-range of the available particles in a bulk is known, the particle size distribution can be determined. This describes the relative amounts of the different-sized particles within a bulk. The distribution of the particle sizes can have a large effect on the properties of a bulk material [11]. When deviations between the particle sizes in a bulk are very small, the particles are monodisperse, but more often than not, the particle sizes vary significantly, and the size distribution is called polydisperse [10].

When dealing with a polydisperse material, it is important to keep in mind that the difference in particle size can have an effect on determining the behaviour and the bulk characteristics of that material [8].

Particle Shape Distribution

Similarly to particle size, particles within a bulk can also have different particle shapes. Particle shape can have an influence on properties like flowability and packing of particles in a bulk [9]. Three different scales can be identified in which particle shape becomes important, the macroscale which is related to the general 3D form of the particles, the mesoscale which is related to the general roundness and angularity of the contour of the particle and the microscale which is related to the smoothness and porosity of the particles [10]. Again, it is important to note that different ways of determining the particle shape distribution exist. When using the particle shape or particle shape distribution, it is important to distinguish the way in which this is measured [9, 10].

Bulk Density

The bulk density is an important characteristic for dry particulate materials, determining the relation between the mass and the volume of a bulk of particles. Because the interstitial space between the particles is included in the bulk density, the packing of the particles is also of importance. A distinction can be made between two types of packing; loose packing, a method of packing particles by dropping these particles from a standardized height through a standardized funnel, or tight packing, a method of packing the particles by tapping the bulk of material with a given mass [10]. The packing of the bulk material has an influence on the interstitial space, or voids, of the bulk. The ratio of the void volume to the total volume of the bulk determines the porosity [12]. Because the bulk density measures the density of a bulk of material while taking into account the density of the solid material and the amount of porosity, it is important to note that the bulk density affects the bulk strength [11].

Bulk Elasticity and Plasticity

A bulk of material can experience elastic or plastic deforma-

tion when subjected to certain loads and stresses. When the material exhibits elastic behaviour, the strain will return to the original point when the stress is released. However, when it exhibits plastic behaviour, a part of the strain will remain after the stress is released. For bulk solids, elastic deformation is often negligible because it does not play a particularly significant role in bulk solid applications. This is due to the fact that the particles will often start moving when the bulk is subjected to stress. This means that the bulk has plastically deformed instead of elastically. Plastic deformations of bulk solids can contribute to changes in material characteristics of the bulk. Plastic deformation of bulk solid material generally refers to the moving of particles with respect to each other, rather than the deformation



Figure 2.2: Stress-strain relationship for elastic (a) and plastic (b) deformation of solids [12].

of individual particles. Common changes in bulk material characteristics due to plastic deformation are for example a change in bulk volume or a change in bulk density. This in turn can have an effect on the stiffness of the bulk of material. What happens with the bulk density and the bulk volume depends on the consolidation of the bulk before it was plastically deformed, but also on the load that is exerted on the bulk that set in motion the plastic deformation [12].

Compressive and Tensile Strength of Bulk

The compressive strength of a material describes the tendency of the material to withstand a crushing or compressive force before the material deforms plastically. For bulk solids, the unconfined yield strength is often used to give a measure of the compressive strength of the bulk. The unconfined yield strength is determined by first compressing the bulk material with a given consolidation stress. Next, the boundaries surrounding the bulk are removed. And finally, the bulk is compressed using a new stress. The stress causing failure is referred to as the unconfined yield strength. However, throughout this work, the term compressive strength is used. The tensile strength of the material describes the tendency of the material to withstand a pulling or tensile force before the material deforms plastically [12].

Internal Friction between Particles

The internal friction between particles can be expressed using the angle of internal friction. It determines the directions of sliding of powder particles in relation to the principle planes of these powder particles [13]. The angle of internal friction is a measure of the force that is required to cause particles to move or slide on each other. This angle is influenced by material characteristics such a particle surface friction, particle shape, particle hardness, particle size and particle size distribution [14]. An important distinction must be made between the angle of internal friction and the effective angle of internal friction. The angle of internal friction is defined as the local slope of the yield locus against the σ -axis. It is often sufficient to construct the angle of internal friction using the linearized yield locus. In this case, the angle of internal friction is determined by the slope of this linearized yield locus with the σ -axis. The



Figure 2.3: Yield locus and Mohr stress circles defining the unconfined yield strength and the consolidation stress [12].

effective angle of internal friction is the slope of the effective yield locus and defines the ratio of the minor principal stress to the major principal stress at a steady state flow. The angle of internal stress usually decreases with an increasing consolidation stress [12]. The angle of internal friction is measured using a specialized shearing box in which the bulk material particles slide on each other [15]. The higher the angle of internal friction, the more difficult it is for the powder or bulk solid material to move [16].

Cohesion of Material

Cohesion is a term that describes the phenomenon of particles in a bulk solid sticking together. It can also be expressed as the resistance of the bulk solid to shear at zero compressive normal load [13]. The cohesion value can be found when looking at the intersection point of the yield locus with the shear axis at the point where the normal stress is equal to zero [12]. A couple of sources of cohesion that can be identified include liquid bridges when moisture is present, van der Waals forces, electrostatic forces and magnetic forces. The first two are the most prevalent forces that cause cohesion between particles. An increase in cohesion can be caused by an increase in moisture content or a decrease of the particle size [14]. For many powders, the value of cohesion can be approximated by taking twice the tensile strength of the bulk material. Additionally, more cohesive bridging is present in a material with a high bulkiness and therefore a high porosity [17]. As was said before, an increase in cohesion can be caused by an increase in cohesion can be caused by an increase in cohesion can be approximated. This is due to the water content between the particles. This water also prohibits the production of new bonds [18].

Flowability or Flow Function

Flowability can be defined as the ability of bulk solid material to flow. It is greatly affected by mechanical characteristics like particle size and moisture content, as well as operational conditions like storage time, temperature, pressure and relative humidity. These material characteristics can cause caking or bridging of the material, which in turn influences the flowability. Flowability itself is not a natural material property, therefore it can be best described using the flow function, also called the failure function of the material [14]. Different consolidation stresses yield different unconfined yield strengths. The instantaneous flow function is a plot of these variables



Figure 2.4: Unconfined yield strength dependent on consolidation stress [12].

Angle of Repose and Shear Angle

The angle of repose describes the slope of a heap of bulk solid material. Various ways of defining an angle of repose can be distinguished. The three main ways are the poured angle of repose, the drained angle of repose and the dynamic angle of repose.



Figure 2.5: Poured angle of repose (a), drained angle of repose (b), dynamic angle of repose (c) [12].

The poured angle of repose is determined by pouring the bulk solid material through a funnel onto a flat surface with a protruding outer edge. This funnel can either be held fixed at a constant height above the flat surface, or moved upwards at a constant height above the top of the heap. This way of determining the angle of repose can only deal with materials at a low compaction grade. This type of compaction is usually present at the surface of a pile of material. It is also important to note that dynamic effects caused by dropping the material from a certain height might influence the results of the angle of repose test, and therefore this angle of repose is not applicable in situations in which greater stresses or time consolidation play a role [12]. When a big angle of repose is measured, it usually signifies that the particles stick together due to cohesion, causing a difficult flow, whereas a small angle of repose signifies a material that flows relatively easy. Usually, the angle of repose increases with an increasing moisture content [14]. Additionally, the angle of repose is influenced by the friction between the particles as well as the cohesion between the particles [19].

The drained angle of repose also describes the slope of a stationary heap of bulk solid material, however the way of determining this drained angle of repose is different compared to the angle of repose. This angle is determined by positioning the bulk solid material in a flat-bottomed box. Subsequently, a central outlet is used to discharge parts of the bulk solid material. The drained angle of repose is then determined by the slope of the material that remains within the container. Because the material is first positioned in a box and later drained through the central outlet, this material is able to consolidate somewhat. Therefore, the stresses in the material that cause this consolidation influence the drained angle of repose. The larger these stresses, the steeper the drained angle of repose shall be [12].

The dynamic angle of repose describes the slope of a dynamically rotating heap of bulk solid material. It is determined by slowly rotating a partly filled cylinder horizontally around its axis and measuring the slope of the material inside the cylinder [12]. The dynamic angle of repose can help to observe the behaviour of the powder when subjected to dynamic surroundings and can indicate some degree of cohesion within the material [13], because cohesive materials do not flow downward continuously, but tend to fall down like avalanches which makes the process of determining the angle of repose much more difficult. The dynamic angle of repose is dependent on the revolving speed, the state of fluidization, the ratio between the particle size and the cylinder size and possible segregation and agglomeration of the material [12].

The shear angle is somewhat of a synonym of the angle of repose, and these two terms are often used interchangeably [7]. Another term that is used to describe the angle of repose is the avalanche angle [20].

2.3. Selection of Relevant Bulk Material Characteristics

A selection is made of the relevant bulk material characteristics listed in Figure 2.1.

Class 2: Bulk material characteristics

- v Particle size distribution: Although the particle size of particles in a bulk of material can vary significantly, the size of the particles can be determined easily using a sieve. Therefore, it is expected that sufficient data is available on the particle size distributions of the materials that have been selected.
- **x** Particle shape distribution: The particle shape distribution has been omitted from this research because a preliminary scan of the available literature showed that not sufficient data was available.
- **v** Bulk density: The dry bulk density is often well documented for granular bulk materials, and therefore it is incorporated in this research.
- Bulk elasticity and plasticity: The bulk elasticity and plasticity are important measures of the behaviour of a material when it is subjected to stress. Therefore, it has to be incorporated in this research.
- v Compressive strength: The compressive strength is an important measure of how well a bulk material can withstand compressive stresses before deforming plastically. Therefore, it must be incorporated in this research.
- v Tensile strength: The tensile strength is an important measure of how well a bulk material can withstand tensile stresses before deforming plastically. Therefore, it must be incorporated in this research.
- **x Internal friction:** The importance of incorporating the internal friction of a bulk material is believed to be covered by the incorporation of the angle of repose and the flowability into this research. Therefore, the internal friction is omitted from this research.
- **v Cohesion:** Cohesion of a bulk material is highly influenced by the presence of humidity, as well as consolidation. Therefore, cohesion is incorporated in this research.
- Flowability or flow function: The flowability of a bulk solid material is dependent on various mechanical characteristics as well as operational conditions. Therefore, it is incorporated in this research.
- v Angle of repose and shear angle: The angle of repose is easy to measure using one of the discussed methods. Therefore, it is expected that sufficient data is available. Therefore, the angle of repose is incorporated in this research.

A further selection has been made of these mechanical characteristics. For the materials selected in section 3.2, numerical values have been assigned to the particle size distribution, bulk density and angle of repose. These have been used to compare the materials with one another in chapter 4. These three mechanical characteristics have been selected because the particle size, bulk density and angle of repose are often measured in laboratory setups, which means that sufficient data can be found, and the dryness of the samples is often known. The other mechanical characteristics selected in section 2.3 have been used to investigate their behaviour when subjected to various operational conditions in chapter 5.

3

Resembling Materials

In this chapter, the chemical properties of sodium borohydride have been investigated and documented. On top of that, a list of twenty inorganic, crystalline, salt-like, day-to-day materials has been created. Last but not least, an overview of the numerical values of the mechanical characteristics discussed in chapter 2 of these materials has been created.

3.1. Analysis of the Chemical Properties of Sodium Borohydride

Sodium borohydride is an inorganic sodium salt. It appears as white to greyish crystals or crystalline powder [21]. In order to find materials with a close resemblance to sodium borohydride, it is important to look for inorganic, crystalline salts.

According to Encyclopaedia Britannica, the field of chemistry defines a salt as the product of a reaction between an acid and a base. The positive ion, or cation, of the base, and the negative ion, or anion, of the acid react together and neutralize each other to form an electrically neutral salt and another product. When a salt is dissolved or in a molten state, most salts will fully disintegrate into their respective positively charged cations and their negatively charged anions again. These are also called electrolytes [22]. The positively charged ions, or cations, are ions that lose electrons in the neutralization reaction. Most metals tend to want to lose their valence electrons, and therefore most metals are cations. The negatively charged ions, or anions, are ions that gain electrons in the neutralization reaction. Most non-metals tend to accept the valence electrons, and therefore they can be classified as anions [23].

Within the category of salts, a distinction can be made between organic salts and inorganic salts. An inorganic salt, is a salt that does not contain carbon-hydrogen bonds, and logically, an organic salt is a salt that does contain carbon-hydrogen bonds [24]. A crystal or crystalline powder is signified as a solid material in which the atoms of that material are arranged in a recurring pattern. The smallest individual part of that recurring pattern is called the unit cell [25].



Figure 3.1: Unit cells for face-centred and body-centred cubic lattices [25].

3.2. Comparable Inorganic, Crystalline, Salt-like Materials

In order to find materials that bear a resemblance to sodium borohydride, and make sure that the materials that are being researched in this assignment microscopically resemble one another, it is important to look for inorganic, crystalline salts. However, most salts have a crystalline structure [26], therefore, to further limit the scope of this research, only materials that have a relative day-to-day use have been researched. This was done due to the fact that presumably, ample research has been done regarding these materials, and the production methods and storage conditions of these materials might be well known and well documented. The twenty selected materials and sodium borohydride have been listed in Table 3.1 below, accompanied by their day-to-day use.

	Material			
Name	Synonym	Code	Application	
Sodium Borohydride	-	NaBH4	Make other chemicals, treat waste water, store hydrogen	[21]
Aluminium Sulfate	-	Al2(SO4)3	Water purification, pH regulation, coagulating agent, deodorant	[27]
Ammonium Nitrate	-	NH4NO3	Fertilizer, explosive in mining industry	[28]
Calcium Carbonate	Limestone/chalk	CaCO3	Construction, paper, plastics, paint	[29]
Calcium Chloride	-	CaCl2	Drying agent, road de-icing, food preservation	[30]
Calcium Sulfate	Gypsum	CaSO4	Building materials, dentistry, cast	[31]
Copper(II) Sulfate	-	CuSO4	Pesticide, germicide, feed additive, soil additive	[32]
Iron(II) Sulfate	-	FeSO4	Water or sewage treatment, fertilizer ingredient	[33]
Magnesium Chloride	-	MgCL2	Medicine, cathartic, alloys,	[34]
Magnesium Sulfate	Epsom Salt	MgSO4	Laxative, fertilizers, bath salts	[35]
Potassium Carbonate	-	K2CO3	Catalyst, fertilizer, flame retardant	[36]
Potassium Chloride	-	KCI	Fertilizer, water softener, production of potassium metal	[37]
Potassium Nitrate	Indian Saltpetre	KNO3	Fertilizer, gunpowder, food preservative, fireworks, toothpaste	[38]
Silicon Dioxide	Sand	SiO2	Sand	[39]
Sodium Bicarbonate	Baking Soda	NaHCO3	Baking agent, antacids, fire extinguisher	[40]
Sodium Carbonate	Washing Soda	Na2CO3	Glass, soap, paper	[41]
Sodium Chloride	Table salt	NaCl	Cooking, preserving food, de-icing agent	[42]
Sodium Metabisulfite	-	Na2S2O5	Food preservative/antioxidant and laboratory agent	[43]
Sodium Nitrate	Chile Saltpeter	NaNO3	Solid propellants, explosives, fertilizers, food preservative	[44]
Sodium Sulfate	-	Na2SO4	Electrolyte replenisher	[45]
Zinc Sulfate	-	ZnSO4	production of rayon, feed supplement, fertilizer ingredient	[46]

Table 3.1: List of the selected materials.

3.3. Mechanical Characteristics of these Materials

An analysis of the mechanical characteristics of the materials discussed in section 3.2 has been conducted. Numerical values have been listed in Table 3.2 for the particle size, the bulk density and the angle of repose. These values have been used in chapter 4 to compare the materials with sodium borohydride.

Material	Particle size [micron]			Bulk density [kg/m^3]			Angle of repose [°]		
	Min	Max		Min	Max		Min	Max	
Sodium Borohydride	1000	4000	[47]	510	530	[47]	34	36	[47]
Aluminium Sulfate	100	2500	[48]		1041	[49]		32	[50][51]
Ammonium Nitrate	100	1000	[52]		785	[49]	30	44	[50][51]
Calcium Carbonate		50	[53]		705	[49]	38	45	[50][54]
Calcium Chloride	500	5000	[55]	817	1057	[49][56]	28	32	[56]
Calcium Sulfate	26	124	[57]		721	[49]	30	42	[50]
Copper(II) Sulfate	150	2000	[58]		833	[49]		31	[50]
Iron(II) Sulfate	220	1000	[59]		1282	[49]			
Magnesium Chloride		500	[60]		192	[49]			
Magnesium Sulfate	200	7000	[61]		833	[49]	30	40	[62]
Potassium Carbonate	100	2500	[63]		1185	[49]	30	40	[62]
Potassium Chloride	500	2000	[64]		961	[49]	30	40	[62]
Potassium Nitrate	50	250	[65]		800	[66]	30	40	[62]
Silicon Dioxide	62	500	[67]		1586	[49]	34	35	[50][54]
Sodium Bicarbonate	65	70	[68]	801	2200	[49][69]	35	42	[50][70]
Sodium Carbonate	75	600	[71]	950	1150	[72]		38	[73]
Sodium Chloride	1	1000	[74]	1282	1330	[49]		25	[50]
Sodium Metabisulfite				1000	1200	[75]			
Sodium Nitrate	519	1223	[76]		1346	[49]		24	[50]
Sodium Sulfate	30	450	[77]		1362	[49]	30	40	[62]
Zinc Sulfate				1281	1442	[78]			

Table 3.2: Particle size range, bulk density and angle of repose of the selected materials.

4

Sodium Borohydride

In the previous chapter, the various materials have been listed, including their particle size, bulk density and angle of repose. In this chapter, these mechanical characteristics have been used to compare the various materials with one another and with sodium borohydride.

4.1. Comparison between Materials and Sodium Borohydride

Particle Size vs. Bulk Density

The ranges of the particle sizes and the bulk densities of the various materials have been plotted in Figure 4.1. Two things can be concluded from this plot. Firstly, the bulk density of sodium borohydride is not similar to the bulk densities of the other materials. This means that when primarily looking at the bulk density, the behaviour of sodium borohydride can not be approximated by any of the selected materials. Secondly, there are a couple of materials that possess a roughly similar particle size range, like calcium chloride and a bit like magnesium sulfate.



Figure 4.1: A plot of the particle sizes vs. the bulk densities.

Particle Size vs. Angle of Repose

The ranges of the particle sizes and the angles of repose of the various materials have been plotted in Figure 4.2. As discussed in section 4.1, the particle size of sodium borohydride is a bit similar to that of calcium chloride and magnesium sulfate. On top of that, the plot of sodium borohydride has some overlap with potassium carbonate and potassium chloride. Additionally, the angle of repose of silicon dioxide and sodium borohydride are quite similar.



Figure 4.2: A plot of the particle sizes vs. angles of repose.

Angle of Repose vs Bulk Density

The ranges of the angles of repose and the bulk densities of the various materials have been plotted in Figure 4.3. As was concluded in section 4.1, the bulk density of sodium borohydride is nowhere near any of the other materials. However, as concluded in Figure 4.1, the angle of repose of sodium borohydride does overlap somewhat with a number of the materials.



Figure 4.3: A plot of the angles of repose vs. the bulk densities.

4.2. Conclusions

Based on the above plots, a couple of things can be concluded. Some of the analysed materials bear a small resemblance to sodium borohydride, and these have been further discussed in the sections below.

Calcium Chloride

Calcium chloride bears a close resemblance to sodium borohydride when looking at the particle size distributions. The ranges of the particle sizes are quite similar, as can be seen in Figure 4.2. The bulk density of calcium chloride, on the other hand, is about twice as high as the bulk density of sodium borohydride, and the angle of repose differs approximately 5 degrees.

Magnesium Sulfate

The plots of magnesium sulfate and sodium chloride overlap in Figure 4.2, although the ranges of magnesium sulfate are quite a lot larger than that of sodium borohydride. However, the bulk density difference between magnesium sulfate and sodium borohydride is quite significant.

Potassium Chloride

Potassium chloride bears a resemblance to sodium borohydride when looking at the particle size distribution. In Figure 4.2 the plots overlap somewhat, signifying that their angles of repose and their particle size are alike. The bulk density however is almost twice as high for potassium chloride compared to sodium borohydride.

Potassium Carbonate

Potassium carbonate bears a resemblance to sodium borohydride when looking at the particle size distribution. In Figure 4.2 the plots overlap somewhat, signifying that their angles of repose and their particle size are alike. The bulk density however is more than twice as high for potassium carbonate compared to sodium borohydride.

Silicon Dioxide

Silicon dioxide resembles sodium borohydride when looking at the angle of repose. Their angles of repose are very similar. However, The bulk density of silicon dioxide is more than three times as high as that of sodium borohydride.

Conclusion

The results of the analysis of the mechanical characteristics of these materials show that although the materials that have been studied all consist of inorganic, crystalline salts, Their respective mechanical characteristics can still differ quite a lot. This means that none of the materials studied in this research sufficiently resemble sodium borohydride.

5

Operational Conditions

Throughout the world, silos are used to store all kinds of bulk solid materials. Because of their height, silos can hold the same amount of material on a smaller surface area when compared to other means of storage. On top of that, because silos are enclosed, the material is somewhat protected from outside influences. Last but not least, by keeping the material in one place, it is easier to control and register the operational conditions influencing the storage conditions of the material [79]. However, the design of the silo is of great importance to the effectiveness of that silo, since a poorly designed silo might lead to periods of shut-down. This might be caused by product degradation, segregation, or a blockage of the flow. And these problems might in turn be caused because the material inside the silo is susceptible to the operational conditions in and around the silo [12]. In this chapter, the operational conditions relevant in bulk storage situations are introduced. Additionally, the influence of the operational conditions on the mechanical characteristics of the materials is discussed.

5.1. Definitions of the Operational Conditions

Operational conditions describe influences on the mechanical characteristics that are independent of the material or the handling equipment. These originate with the operational conditions to which the material and equipment are subjected during handling and storage. A couple of different operational conditions important for the analysis of the storage of a material can be identified. These operational conditions are temperature, relative humidity, and stress history, which can be subdivided into handling types and time consolidation [5][6][7]. These operational conditions can influence the material and its behaviour during bulk handling and storage greatly, and they can also influence the various material characteristics that have been discussed in section 2.1.



Figure 5.1: Overview of the operational conditions.

Temperature

Temperature is an obvious operational condition influencing the storage conditions of bulk materials. Throughout the day, temperatures can vary quite significantly. This temperature range is subject to conditions like the season, the number of sunlight hours, the cloudiness, the elevation of the location where the silo is placed, the humidity, and the wind speed [80]. The temperature at the location of storage can have a significant influence on the flowability of the material that is being stored. The moisture present in the material might freeze, impacting the flowability. It is also possible that the

material undergoes a change in crystallinity due to temperature changes. On top of that, other physical properties can also change due to temperature changes, causing a difference in flow and behaviour of the material [12][14].

Relative Humidity

Relative humidity (RH) describes a ratio of the atmospheric moisture relative to the saturated atmospheric moisture level. The saturated atmospheric moisture level is dependent on temperature, therefore, the relative humidity is dependent on both the temperature and the moisture content [81]. The relative humidity inside a silo, as well as in the interstitial spaces of the material, can also affect the bulk material properties. Many materials take up moisture when subjected to humid conditions, which in turn can change the mechanical characteristics of the material [14].

Stress History

The stress history describes phenomenons like consolidation stress, exerted on the bulk leading to possible adhesion, as well as other changes in mechanical characteristics [12]. These stresses can be caused by a variety of factors, such as the ways in which the material is handled, as well as the time that the material has been stored. The bulk of material weighs on itself, causing a stress. Vibrations due to loading and unloading might also impact the stresses exerted on the bulk material. As a result of this stress, a larger number of contact points between the particles can be distinguished, which in turn can lead to an increased interparticle adhesion [14].

5.2. Influence of the Operational Conditions

It can be expected that the mechanical characteristics of every individual material react differently to the influence of these operational conditions. In Table 5.1, the effects of bulk storage operational conditions on the mechanical characteristics of certain materials as described in literature is presented. As can be seen in Table 5.1, only significant data has been reported on sodium chloride, calcium carbonate and sodium bicarbonate and therefore the other materials that have been identified in chapter 3 have been omitted.

In this table, the colours green, red, yellow, and gray have been used to signify an effect of that specific operational condition on that specific mechanical characteristic of that specific material. Green signifies that when the operational condition increases, the mechanical characteristic increases as well. Red means that when the operational condition increases, the mechanical characteristic decreases. Yellow means that an increase in the operational condition might first increase the mechanical characteristic and later decrease it, or the other way around. Gray means that increasing or decreasing the operational condition does not have an effect on the mechanical characteristic of the material, but it rather stays relatively constant. In sections 5.2.1, 5.2.2 and 5.2.3, these findings have been described in further detail.

Material	PS	BD	BE	CS	TS	AoR	С	F			
Temperature											
Sodium Chloride											
Calcium Carbonate				[82]	[82]		[82]				
Sodium Bicarbonate	[83]			[82]	[82]		[82]				
Relative Humidity											
Sodium Chloride	[84]	[20]	[85]	[85][86][87]	[86][87][88][89][90]	[14][20]	[84]	[14][84][20][87][90]			
Calcium Carbonate				[82]				[91]			
Sodium Bicarbonate	[83]			[82]	[86]		[83][86]				
Stress											
Sodium Chloride		[20]		[85][87]	[87][90][92]	[20]	[93]	[20][87][90]			
Calcium Carbonate		[94]		[82]			[82]				
Sodium Bicarbonate	[95]			[82]			[82]				



Sodium Chloride

Temperature

 No relevant information could be found on the influence of temperature on the various mechanical characteristics of Sodium Chloride.

Relative Humidity

- Particle size distribution: Relative humidity clearly does have an effect on the mechanical characteristics of sodium chloride. Increasing humidity causes particles to take up moisture. Especially smaller crystals tend to absorb moisture during storage due to the relatively higher surface area compared to volume. With a drop in humidity, evaporation causes recrystallizing of the particles. Sometimes the bonds between these particles are so strong, that the barrier between the particles becomes invalid, creating new, bigger particles [84].
- Bulk density: An increased relative humidity has a predictable effect on the bulk density of sodium chloride, given the fact that the moisture content of sodium chloride increases when subjected to a higher relative humidity. This can be explained because the moisture takes up the interstitial spaces, whereas before there used to be air in these spaces [20].
- Bulk elasticity and plasticity An increased relative humidity leads to an increased moisture content in sodium chloride, which in turn has a positive effect on the crystal plasticity [85].
- Compressive strength: The increase in moisture content due to a higher relative humidity causes sodium chloride to have a higher compressive strength, due to an increase in interparticle forces caused by the hydrogen bridges [85, 86, 87].
- Tensile strength: Similar to the compressive strength, the tensile strength is increased with a higher relative humidity due to a higher moisture content, and in turn due to an increase in interparticle forces caused by the liquid bridges [86, 87, 88, 89, 90].
- Angle of repose: Sodium chloride stored at lower relative humidity showed a lower angle of repose than material stored at higher relative humidity [14, 20].
- Cohesion: An increase in relative humidity causes an increase in moisture content, which in turn causes more liquid bridges and an increase in interparticle cohesion [84].
- Flowability: An increase in relative humidity during storage decreases the flowability of sodium chloride due to an increased interparticle attraction. On top of that, an increased relative humidity causes an increased angle of repose, which in turn signifies a decreasing flowability. In the same way, an increased level of cohesion between the particles shows a decrease in flowability [14, 84, 20, 87, 90].

Stress History

- Bulk density: As a result of compaction, the bulk density of sodium chloride increases over time. This is due to the fact that the material 'settles' and the voids in between the particles become smaller. Therefore, the total volume used to determine the bulk density becomes smaller and the bulk density increases [20].
- Compressive strength: An increase in the compressive or crushing strength of sodium chloride has been documented. This is a time dependent increase, where the compressive strength increases with an increasing storage time. It must be said however that this increase in strength is only caused when humidity is present. The compaction forces of the time consolidation make sure the particles are sufficiently close to one another in order for the humidity to cause the formation of liquid bridging between the particles [85, 87].
- Tensile strength: For the tensile strength, it has been documented that an increased packing density causes an increase in tensile strength of sodium chloride. As shown before, this increased density can be caused by an increase in humidity, but also by an increase in compaction. Time consolidation causes this compaction in storage, making for an increased tensile strength of the material [87, 90, 92].
- Angle of repose: An increasing storage time has been shown to increase the angle of repose
 of sodium chloride. This is due to the compaction of the material caused by its own weight. This
 compaction of the material increases over time [20].
- Cohesion: Initially, crystalline materials like sodium chloride do not show caking or cohesive effects due to time consolidation. This might however happen when the relative humidity around the sodium chloride exceeds a certain level. At that point, the surface of the particle might dissolve.

When these particles are subjected to stress, possibly due to consolidation, these layers of dissolved material can touch and the material can form strong connections upon drying. Cohesion between the particles has occurred [93].

- Flowability: Material flow is impacted by the compaction of the material. This compaction is caused by time consolidation, where the material weighs on itself. This decreases the flowability of the material [20, 87, 90].

Calcium Carbonate

Temperature

- **Compressive strength:** No significant change in the strength of the calcium carbonate material has been observed when exposed to an increase in temperature [82].
- **Tensile strength:** The tensile strength of calcium carbonate agglomerates tends to increase with a higher drying temperature [82].
- Cohesion: Given that moisture is present in the initial sample of calcium carbonate, an increase in temperature causes an increase in the formation of agglomerates, or in other words, an increase in cohesion between the particles. This is because a higher drying temperature would cause the moisture to evaporate faster and the material to dry faster. This in turn would mean that the crystals would be less exposed to shearing during this drying. Therefore, the chance for the agglomerates to break again would decrease, meaning an increase in the cohesion between the particles [82].

Relative Humidity

- Compressive strength: The strength of the calcium carbonate agglomerates increased with a
 decreasing moisture content. An increasing relative humidity causes an increase in moisture
 content in calcium carbonate and therefore a decrease in agglomerate strength [82].
- Flowability: For small particles (1 micron), moisture causes agglomeration. In turn, the material forms tiny agglomerated balls that have an increasing effect on the flowability of the material. For larger particles (15 micron), moisture caused the flowability to decrease [91].

Stress History

- Bulk density: For calcium carbonate, consolidation stress has a large effect on the bulk density. An increase in consolidation stress causes an increase in the material bulk density [94].
- Compressive strength: The strength of calcium carbonate agglomerates is independent of the consolidation stress caused by its own weight weighing down on it, however, agitation stress caused an increase in strength of the calcium carbonate agglomerates [82].
- Cohesion: An increase in agitation stress causes the cohesion between the calcium carbonate particles to increase. This is due to the enhancement of the compression and consolidation processes surrounding the calcium carbonate particles, which in turn leads to stronger agglomerates [82].

Sodium Bicarbonate

Temperature

- Particle size distribution: The size distribution after drying the samples using microwave irradiation stayed the same, however it is important to note that sodium bicarbonate degrades into sodium carbonate when subjected to high temperatures. This is due to the evaporation of the structural water inside the material. By removing that from the material, the sodium bicarbonate decomposes into sodium carbonate [83].
- Compressive strength: An increase in temperature leads to a decrease of sodium bicarbonate agglomerate strength. Although the material shows an increase in the amount of agglomerates, they are of brittle nature and their compressive strength decreases with an increasing temperature. However, the overall crush strength of the material increases due to stronger interparticle connections because of the recrystallization of the material at the higher drying temperatures [82].
- **Tensile strength:** An increasing drying temperature might cause the formation of agglomerates with an increased tensile strength [82].

Cohesion: An increase in temperature causes an increase in the formation of agglomerates. In
other words, an increase in temperature increases the amount of cohesion between the particles
of sodium bicarbonate [82].

Relative Humidity

- Particle size distribution: The average particle size and the particle size distribution showed no significant changes due to a decrease in moisture content. It may be expected that an increase in relative humidity, and consequently moisture content, of the sodium bicarbonate particles shall not influence the particle size distribution [83].
- Compressive strength: At a higher temperature, the solubility of sodium bicarbonate increases. If the relative humidity is high, more moisture shall be taken into the material sample, in turn leading to stronger connections between the individual crystals [82].
- Tensile strength: The tensile strength of sodium bicarbonate first increases with an increasing relative humidity, but at an intermediate relative humidity, a maximum in tensile strength can be identified, after which the tensile strength starts to decrease again with an increasing relative humidity [86].
- Cohesion: An increase in relative humidity causes an increase in moisture content, which in turn causes an increase in the cohesiveness of sodium bicarbonate [83, 86].

Stress History

- Particle size distribution: The stresses caused by an increasing compaction stress cause a so-called brittle fracture, which causes the particles to degrade and become smaller. So, an increasing compaction stress causes a decreasing particle size [95]
- Compressive strength: An increase in the agitation speed caused a decrease in agglomerate strength, however, the height of the material bed did not have an effect on the strength of the sodium bicarbonate agglomerates [82].
- Cohesion: A decrease in material bed height caused an increase in the amount of agglomerates that were formed. In other words, a decrease in consolidation stress caused an increase in the cohesiveness between the sodium bicarbonate particles [82].

5.3. Conclusions

In section 4.1, five materials have been identified that somewhat resemble sodium borohydride. These were calcium chloride, magnesium sulfate, potassium chloride, potassium carbonate and silicon dioxide. In section 5.2, the influence of the operational conditions has been documented on sodium chloride, calcium carbonate and sodium bicarbonate. None of these three materials can be matched to sodium borohydride using the mechanical characteristics. The result of this literature research shows that al-though the materials that have been studied all consist of inorganic, crystalline salts, the effect that the operational conditions have on the mechanical characteristics of the materials can still vary significantly. Given the limited amount of data available in the scientific literature concerning itself with the effect of operational conditions on the mechanical characteristics of certain materials, it is proposed to increase the effort in obtaining information on the effect of operational conditions by using an experimental setup in a laboratory, rather than commencing another literature research. It is proposed to look further into the relation between sodium borohydride and materials like calcium chloride, magnesium sulfate, potassium chloride, potassium carbonate and silicon dioxide.

Conclusion

Throughout this literature research report, the pursuit of knowledge about the operational conditions influencing the mechanical characteristics of various inorganic, crystalline salts has been highlighted. The purpose of this research was to make a prediction on the effect of these operational conditions on the behaviour of sodium borohydride, by comparing sodium borohydride to these other inorganic, crystalline salts.

A breakdown of the well-known mechanical characteristics surrounding the classification of material behaviour has provided a sturdy framework in which this literature research has been conducted. The selected bulk material characteristics; particle size distribution; bulk density; bulk elasticity and plasticity; compressive strength; tensile strength; angle of repose; cohesion and flowability have been introduced in order to provide the reader with a definition to uphold throughout the text.

An analysis of the chemical properties of sodium borohydride helped to guide this research towards a list of relevant, inorganic, crystalline salts. On top of that, another selection criterium used to select these materials was the need to have a significant day-to-day use, with the hope that this would mean an abundance of information of the production and storage of these materials would be available. An analysis of the particle sizes, bulk densities and angles of repose of these materials was conducted in order to construct a framework. This framework could later be used to compare sodium borohydride to the other materials.

An effort was made to compare sodium borohydride to the other materials. Using the particle sizes, bulk densities and angles of repose, sodium borohydride could be linked somewhat to calcium chloride, magnesium sulfate, potassium chloride, potassium carbonate and silicon dioxide. It is however proposed to commence with additional laboratory research to try to match these materials to one another more thoroughly.

The three main operational conditions; temperature; relative humidity and stress history, have been introduced. Only significant data has been reported on sodium chloride, calcium carbonate and sodium bicarbonate. Although the influence of the various operational conditions does have an effect on the selected mechanical characteristics of these materials, no definitive conclusions could be drawn about the effect of the operational conditions on the mechanical characteristics.

For further research, it is proposed to invest more time in the effort to match various materials to sodium borohydride. This could be done on the basis of mechanical characteristics as is done in this research, however, this could also be done using the chemical properties of the materials to match the materials, rather than the mechanical characteristics. Furthermore, it is proposed to commence with laboratory research to try to investigate the effect of the operational conditions on sodium borohydride directly, by subjecting a bulk of sodium borohydride in storage to the various operational conditions while measuring the various mechanical characteristics.

References

- [1] A. Pariona. Causes of Global Warming. 2017. URL: https://www.worldatlas.com/articles/ causes-of-global-warming-list.html (visited on 12/20/2022).
- [2] N. van Mead. The hydrogen hype: a bubble or a game changer in decarbonization? 2022. URL: https://journeytozerostories.neste.com/innovation/hydrogen-hype-bubble-or-gamechanger-decarbonization (visited on 12/20/2022).
- [3] R.M. Mohring and Y. Wu. "Hydrogen Generation via Sodium Borohydride". In: *AIP Conference Proceedings* (2003). http://dx.doi.org/10.1063/1.1597360.
- [4] Pirelli. HYDROGEN AS A FUEL: THE PROS AND CONS. 2022. URL: https://www.pirelli. com/global/en-ww/road/hydrogen-as-a-fuel-the-pros-and-cons (visited on 12/20/2022).
- [5] D.L. Schott. Lecture slides. Sept. 2021.
- [6] D.L. Schott. Lecture slides. Feb. 2022.
- [7] M.C. van Benten. Personal communication. Dec. 2022.
- [8] K.J. Berger and C.M. Hrenya. "Challenges of DEM: II. Wide particle size distributions". In: Powder Technology 264 (2014). https://doi.org/10.1016/j.powtec.2014.04.096, pp. 627– 633.
- [9] T. Allen. Particle Size Measurement. 3rd ed. Springer, 1981. ISBN: 978-0-412-15410-2.
- [10] H.G. Merkus. Particle Size Measurements; Fundamentals, Practice, Quality. Pijnacker, the Netherlands: Springer, 2009. ISBN: 978-1-4020-9015-8.
- [11] P. Hazelton and B. Murphy. *Interpreting Soil Test Results: What Do All The Numbers Mean?* 3rd ed. CSIRO Publishing, 2017. ISBN: 9781486303960.
- [12] D. Schulze. Powders and Bulk Solids; Behavior, Characterization, Storage and Flow. Springer, 2008. ISBN: 978-3-540-73767-4.
- [13] R.L. Brown and J.C. Richards. *Principles of Powder Mechanics*. 1st ed. Pergamon Press, 1970. ISBN: 9781483145655.
- [14] V. Ganesan, Kurt A. Rosentrater, and K. Muthukumarappan. "Flowability and Handling Characteristics of Bulk Solids and Powders A Review". In: 2005 Tampa, FL July 17-20, 2005. 2005
 Tampa, FL July 17-20, 2005. American Society of Agricultural and Biological Engineers, 2005. DOI: 10.13031/2013.19548. URL: http://elibrary.asabe.org/abstract.asp?JID=5&AID=19548&CID=tfl2005&T=1 (visited on 02/09/2023).
- [15] B.K. Bala. Drying and Storage of Cereal Grains. 2nd ed. Jessore, Bangladesh: Wiley Blackwell, 2017. ISBN: 9781119124238.
- [16] A. Manickavasagan, C. Santahkumar, and N. Venkatachalapathy. *Brown Rice*. Springer, 2017. ISBN: 9783319590103.
- [17] T. Eaves and T.M. Jones. "Cohesion and tensile strength of bulk solids". In: *Rheologica Acta* 10 (1971). https://doi.org/10.1007/BF01972489, pp. 127–134.
- [18] T. Eaves and T.M. Jones. "Effect of Moisture on Tensile Strength of Bulk Solids I: Sodium Chloride and Effect of Particle Size". In: *Journal of Pharmaceutical Sciences* 61.2 (1972). https: //doi.org/10.1002/jps.2600610226, pp. 256–261.
- [19] Libretexts Engineering. 33.4: Angle of Repose. 2020. URL: https://eng.libretexts.org/ Bookshelves/Materials_Science/TLP_Library_I/33%3A_Granular_Materials/33.4%3A_ Angle_of_Repose (visited on 02/22/2023).

- [20] Adam M. Stoklosa et al. "Effects of storage conditions, formulation, and particle size on moisture sorption and flowability of powders: A study of deliquescent ingredient blends". In: Food Research International 49.2 (Dec. 2012), pp. 783–791. ISSN: 09639969. DOI: 10.1016/j. foodres.2012.09.034. URL: https://linkinghub.elsevier.com/retrieve/pii/S09639969 12003936 (visited on 04/04/2023).
- [21] PubChem Sodium borohydride. Sodium borohydride. 2023. URL: https://pubchem.ncbi.nlm. nih.gov/compound/sodium-borohydride (visited on 05/26/2023).
- [22] Encyclopaedia Britannica. Salt acid-base reactions. 2023. URL: https://www.britannica. com/science/salt-acid-base-reactions (visited on 01/11/2023).
- [23] M. Ghadban. What is Salt in Chemistry. 2021. URL: https://study.com/learn/lesson/whatis-salt-in-chemistry-properties-formula.html (visited on 01/13/2023).
- [24] Biology Online. Inorganic Salt. 2021. URL: https://www.biologyonline.com/dictionary/ inorganic-salt (visited on 05/26/2023).
- [25] G.D. Mahan. Crystal. 2023. URL: https://www.britannica.com/science/crystal/Structu re (visited on 05/26/2023).
- [26] Saltworks. About salt. 2022. URL: https://seasalt.com/salt-101/about-salt (visited on 03/02/2023).
- [27] PubChem Aluminum Sulfate. Aluminum Sulfate. 2023. URL: https://pubchem.ncbi.nlm.nih. gov/compound/aluminum-sulfate (visited on 05/26/2023).
- [28] PubChem Ammonium nitrate. Ammonium nitrate. 2023. URL: https://pubchem.ncbi.nlm. nih.gov/compound/Ammonium-nitrate (visited on 05/26/2023).
- [29] PubChem Calcium Carbonate. Calcium Carbonate. 2023. URL: https://pubchem.ncbi.nlm. nih.gov/compound/Calcium-carbonate (visited on 05/26/2023).
- [30] PubChem Calcium Chloride. Calcium Chloride. 2023. URL: https://pubchem.ncbi.nlm.nih. gov/compound/Calcium-chloride (visited on 05/26/2023).
- [31] PubChem Calcium sulfate. Calcium sulfate. 2023. URL: https://pubchem.ncbi.nlm.nih. gov/compound/calcium-sulfate (visited on 05/26/2023).
- [32] PubChem Cupric Sulfate. Cupric Sulfate. 2023. URL: https://pubchem.ncbi.nlm.nih.gov/ compound/copper-sulfate (visited on 05/26/2023).
- [33] PubChem Ferrous Sulfate. Ferrous Sulfate. 2023. URL: https://pubchem.ncbi.nlm.nih. gov/compound/iron-sulfate (visited on 05/26/2023).
- [34] PubChem Magnesium Chloride. Magnesium Chloride. 2023. URL: https://pubchem.ncbi. nlm.nih.gov/compound/Magnesium-chloride (visited on 05/26/2023).
- [35] PubChem Magnesium Sulfate. Magnesium Sulfate. 2023. URL: https://pubchem.ncbi.nlm. nih.gov/compound/Magnesium-sulfate (visited on 05/26/2023).
- [36] PubChem Potassium Carbonate. *Potassium Carbonate*. 2023. URL: https://pubchem.ncbi. nlm.nih.gov/compound/Potassium-Carbonate (visited on 05/26/2023).
- [37] PubChem Potassium Chloride. *Potassium Chloride*. 2023. URL: https://pubchem.ncbi.nlm. nih.gov/compound/Potassium-chloride (visited on 05/26/2023).
- [38] PubChem Potassium Nitrate. Potassium Nitrate. 2023. URL: https://pubchem.ncbi.nlm.nih. gov/compound/Potassium-nitrate (visited on 05/26/2023).
- [39] PubChem Silicon Dioxide. Silicon Dioxide. 2023. URL: https://pubchem.ncbi.nlm.nih.gov/ compound/silicon-dioxide (visited on 05/26/2023).
- [40] PubChem Sodium Bicarbonate. Sodium Bicarbonate. 2023. URL: https://pubchem.ncbi.nlm. nih.gov/compound/Sodium-bicarbonate (visited on 05/26/2023).
- [41] PubChem Sodium Carbonate. Sodium Carbonate. 2023. URL: https://pubchem.ncbi.nlm. nih.gov/compound/Sodium-carbonate (visited on 05/26/2023).
- [42] PubChem Sodium chloride. Sodium Chloride. 2023. URL: https://pubchem.ncbi.nlm.nih. gov/compound/sodium-chloride (visited on 05/26/2023).

- [43] PubChem Sodium Metabisulfite. Sodium Metabisulfite. 2023. URL: https://pubchem.ncbi. nlm.nih.gov/compound/sodium-metabisulfite (visited on 05/26/2023).
- [44] PubChem Sodium Nitrate. Sodium Nitrate. 2023. URL: https://pubchem.ncbi.nlm.nih.gov/ compound/sodium-Nitrate (visited on 05/26/2023).
- [45] PubChem Sodium Sulfate. Sodium Sulfate. 2023. URL: https://pubchem.ncbi.nlm.nih.gov/ compound/sodium-Sulfate (visited on 05/26/2023).
- [46] PubChem Zinc Sulfate. Zinc Sulfate. 2023. URL: https://pubchem.ncbi.nlm.nih.gov/ compound/zinc-sulfate (visited on 05/26/2023).
- [47] M.C. van Benten. Personal communication. June 2023.
- [48] hydro-land. Aluminium sulfate. URL: http://hydro-land.com/e/ligne-en/doc/Aluminium-Sulfate.html#:~:text=It%20is%20available%20in%20various,or%203%20and%2035%20mm (visited on 06/16/2023).
- [49] Anval. Bulk Density Chart. 2023. URL: https://www.anval.net/Downloads/Bulk%20Density% 20Chart.pdf (visited on 05/28/2023).
- [50] ckit. Material Characteristics. URL: https://www.ckit.co.za/secure/misc/trajectories_ material.html (visited on 06/16/2023).
- [51] Scribd. Angle of Repose. URL: https://www.scribd.com/document/434942361/Angle-of-Repose (visited on 06/16/2023).
- [52] Filippo Maggi, L Arosio, and Luciano Galfetti. "Catalyst screening for ammonium nitrate oxidizer". In: July 2017. DOI: 10.13009/EUCASS2017-520.
- [53] Merck. Calcium carbonate. 2023. URL: https://www.sigmaaldrich.com/NL/en/product/ aldrich/310034 (visited on 06/16/2023).
- [54] O. S. Baghabra Al-Amoudi H. M. Beakawi Al-Hashemi. "A review on the angle of repose of granular materials". In: *Powder Technology* 330 (2018), pp. 397–417. DOI: https://doi.org/ 10.1016/j.powtec.2018.02.003.
- [55] Roth. Calcium chloride, 1kg. URL: https://www.carlroth.com/com/en/renewable-desiccan ts/calcium-chloride/p/cn92.2 (visited on 06/16/2023).
- [56] DOW Chemical Company. Calcium Chloride Handbook. 2003. URL: https://www.osi-unive rs.org/IMG/pdf/CalciumChloridHandbook-2.pdf (visited on 06/16/2023).
- [57] Echa. Calcium sulfate. URL: https://echa.europa.eu/nl/registration-dossier/-/regis tered-dossier/15878/4/6 (visited on 06/16/2023).
- [58] Copper Development Association Inc. Copper sulphate. 2013. URL: https://www.copper. org/resources/properties/compounds/copper_sulfate01.html (visited on 06/16/2023).
- [59] A. Arce and Z. Ustunol. "Effect of microencapsulated ferrous sulfate particle size on Cheddar cheese composition and quality". In: *Journal of Dairy Science* 101.8 (2018), pp. 6814–6822. DOI: https://doi.org/10.3168/jds.2017-13952.
- [60] Prachya Rojanotaikul et al. "Preparation of Porous Anhydrous MgCl 2 Particles by Spray Drying Process". In: *Engineering Journal* 16 (July 2012). DOI: 10.4186/ej.2012.16.3.109.
- [61] Sure Chemical Co. *Magnesium sulphate*. 2009. URL: https://www.surechemical.com/other chemicals/Magnesium%20Sulphate.html (visited on 06/16/2023).
- [62] Omega. MATERIAL CHARACTERISTIC GUIDE. URL: https://www.omega.co.uk/green/ pdf/MaterialChar_Guide.pdf (visited on 06/16/2023).
- [63] Sirius. Product Data Sheet Potassium carbonate. 2019. URL: https://sirius-es.com/wpcontent/uploads/2021/06/Potassium-carbonate-PDS-EN.pdf (visited on 06/16/2023).
- [64] P. Loganathan, M. J. Hedley, and S. A. Clark. "The manufacture and evaluation of granular potassium chloride fertilizers". In: *Fertilizer research* 31 (1992), pp. 291–304. DOI: https:// doi.org/10.1007/BF01051281.

- [65] Oleg Linnikov et al. "Kinetics and Mechanism of Spontaneous Crystallization of Potassium Nitrate from Its Supersaturated Aqueous Potassium Nitrate from Its Supersaturated Aqueous Solutions". In: Crystal Structure Theory and Applications 2 (Mar. 2013), pp. 16–27. DOI: 10.4236/ csta.2013.21003.
- [66] Merck. Potassium Nitrate. 2023. URL: https://www.merckmillipore.com/NL/en/product/ Potassium-nitrate, MDA_CHEM-105061 (visited on 05/28/2023).
- [67] Engineering Toolbox. Particle Size. 2005. URL: https://www.engineeringtoolbox.com/ particle-sizes-d_934.html (visited on 06/16/2023).
- [68] D. S. Peterson. Soda Blast Question Revisited. 2012. URL: https://www.pfonline.com/ articles/soda-blast-question-revisited (visited on 06/16/2023).
- [69] Beroil Energy. Review of sodium bicarbonate (baking soda) particle size and its effect based on the method of production. 2022. URL: https://beroilenergy.com/review-of-sodiumbicarbonate-baking-soda-particle-size-and-its-effect-based-on-the-method-ofproduction (visited on 06/16/2023).
- [70] Scribd. Angle of Repose. URL: https://www.scribd.com/doc/233226178/Angle-of-Repose# (visited on 06/16/2023).
- [71] Eti Soda. Sodium Carbonate (Dense Soda Ash). 2020. URL: https://www.etisoda.com/en/ sodium-carbonate-dense-soda-ash/ (visited on 06/16/2023).
- [72] Merck. Sodium Carbonate. 2023. URL: https://www.merckmillipore.com/NL/en/product/ Sodium-carbonate, MDA_CHEM-102405 (visited on 05/28/2023).
- [73] TATA Chemical. Sales Specification. 2011. URL: https://tatachemicalseurope.com/sites/ default/files/specification_Sodium.pdf (visited on 06/16/2023).
- [74] Eti Soda. BUY SALT PRODUCTS WITH CONFIDENCE. 2023. URL: https://www.puresalt. co.uk/buy-salt/ (visited on 06/16/2023).
- [75] Merck. Sodium Disulfite. 2023. URL: https://www.merckmillipore.com/NL/en/product/ Sodium-disulfite, MDA_CHEM-106357 (visited on 05/28/2023).
- [76] NIST. SAMPLE CONDITIONS and PHYSICAL PROPERTIES. 2017. URL: https://webbook. nist.gov/chemistry/silmarils-solids-hrf-drf/docs/META.MIR.PUR.Sodium-nitrate. SLD.SIGMA-ALDRICH.PNNL509421.pdf (visited on 06/16/2023).
- [77] NIST. SAMPLE CONDITIONS and PHYSICAL PROPERTIES. 2017. URL: https://webbook. nist.gov/chemistry/silmarils-solids-hrf-drf/docs/META.MIR.PUR.Sodium-sulfateanhydrous.SLD.SIGMA-ALDRICH.PNNL509893.pdf (visited on 06/16/2023).
- [78] Pestell. Zinc Sulphate. 2023. URL: https://pestell.com/product/zinc-sulphate-35-percent-monohydrate-maxi-granular/ (visited on 05/28/2023).
- [79] Advantec. Pros and Cons of Grain Storage Silos. 2023. URL: https://advantecaustralasia. wordpress.com/2017/04/12/pros-and-cons-of-grain-storage-silos/ (visited on 05/29/2023).
- [80] T. Means. Understanding Diurnal Temperature Range. 2019. URL: https://www.thoughtco. com/diurnal-temperature-range-3444244 (visited on 05/30/2023).
- [81] Wheather.gov. Relative Humidity (RH). URL: https://graphical.weather.gov/definitions/ defineRH.html (visited on 05/30/2023).
- [82] Hong Lee Lim, Karen P. Hapgood, and Brian Haig. "Understanding and preventing agglomeration in a filter drying process". In: *Powder Technology* 300 (Oct. 2016), pp. 146–156. ISSN: 00325910. DOI: 10.1016/j.powtec.2016.03.003. URL: https://linkinghub.elsevier. com/retrieve/pii/S0032591016300961 (visited on 04/06/2023).
- [83] Abdullah Seyrankaya. "Drying of Sodium Bicarbonate with Microwave Energy". In: Asian J. Chem. ().

- [84] Susan R. Feldman. "Sodium Chloride". In: Kirk-Othmer Encyclopedia of Chemical Technology. Ed. by John Wiley & Sons, Inc. Hoboken, NJ, USA: John Wiley & Sons, Inc., Oct. 14, 2011, 1915040902051820.a01.pub3. ISBN: 978-0-471-23896-6. DOI: 10.1002/0471238961.191504 0902051820.a01.pub3. URL: https://onlinelibrary.wiley.com/doi/10.1002/0471238961. 1915040902051820.a01.pub3 (visited on 04/04/2023).
- [85] G.R.B. Down and J.N. McMullen. "The effect of interparticulate friction and moisture on the crushing strength of sodium chloride compacts". In: *Powder Technology* 42.2 (May 1985), pp. 169–174. ISSN: 00325910. DOI: 10.1016/0032-5910(85)80050-4. URL: https://linkinghub.elsevier.com/retrieve/pii/0032591085800504 (visited on 04/04/2023).
- [86] Claes Ahlneck and Göran Alderborn. "Moisture adsorption and tabletting. I. Effect on volume reduction properties and tablet strength for some crystalline materials". In: *International Journal of Pharmaceutics* 54.2 (Sept. 1989), pp. 131–141. ISSN: 03785173. DOI: 10.1016/0378-5173 (89)90332-3. URL: https://linkinghub.elsevier.com/retrieve/pii/0378517389903323 (visited on 04/05/2023).
- [87] Shabbir Dawoodbhai and Christopher T. Rhodes. "The Effect of Moisture on Powder Flow and on Compaction and Physical Stability of Tablets". In: *Drug Development and Industrial Pharmacy* 15.10 (Jan. 1989), pp. 1577–1600. ISSN: 0363-9045, 1520-5762. DOI: 10.3109/036390489090 52504. URL: http://www.tandfonline.com/doi/full/10.3109/03639048909052504 (visited on 04/05/2023).
- [88] Terence Eaves and Trevor M. Jones. "Effect of Moisture on Tensile Strength of Bulk Solids I: Sodium Chloride and Effect of Particle Size". In: *Journal of Pharmaceutical Sciences* 61.2 (Feb. 1972), pp. 256–261. ISSN: 00223549. DOI: 10.1002/jps.2600610226. URL: https: //linkinghub.elsevier.com/retrieve/pii/S0022354915383179 (visited on 02/09/2023).
- [89] Claes Ahlneck and Göran Alderborn. "Moisture adsorption and tabletting. II. The effect on tensile strength and air permeability of the relative humidity during storage of tablets of 3 crystalline materials". In: *International Journal of Pharmaceutics* 56.2 (Nov. 1989), pp. 143–150. ISSN: 03785173. DOI: 10.1016/0378-5173(89)90007-0. URL: https://linkinghub.elsevier. com/retrieve/pii/0378517389900070 (visited on 04/05/2023).
- [90] M. Eriksson and G. Alderborn. "Mechanisms for post-compaction changes in tensile strength of sodium chloride compacts prepared from particles of different dimensions". In: *International Journal of Pharmaceutics* 109.1 (Aug. 1994), pp. 59–72. ISSN: 03785173. DOI: 10.1016/0378-5173(94)90121-X. URL: https://linkinghub.elsevier.com/retrieve/pii/0378517394901 21X (visited on 04/05/2023).
- [91] Brian H. Kaye. "Characterizing the Flowability of a Powder Using the Concepts of Fractal Geometry and Chaos Theory". In: *Particle & Particle Systems Characterization* 14.2 (Apr. 1997), pp. 53–66. ISSN: 09340866. DOI: 10.1002/ppsc.199700013. URL: https://onlinelibrary. wiley.com/doi/10.1002/ppsc.199700013 (visited on 04/26/2023).
- [92] Göran Alderborn and Claes Ahlneck. "Moisture adsorption and tabletting. III. Effect on tablet strenght-post compaction storage time profiles". In: *International Journal of Pharmaceutics* 73.3 (July 1991), pp. 249–258. ISSN: 03785173. DOI: 10.1016/0378-5173(91)90417-M. URL: htt ps://linkinghub.elsevier.com/retrieve/pii/037851739190417M (visited on 04/05/2023).
- [93] Markus Hartmann and Stefan Palzer. "Caking of amorphous powders Material aspects, modelling and applications". In: *Powder Technology* 206.1 (Jan. 2011), pp. 112–121. ISSN: 00325910. DOI: 10.1016/j.powtec.2010.04.014. URL: https://linkinghub.elsevier.com/retrieve/pii/S0032591010001968 (visited on 04/06/2023).
- [94] Hamid Salehi, Diego Barletta, and Massimo Poletto. "A comparison between powder flow property testers". In: *Particuology* 32 (June 2017), pp. 10–20. ISSN: 16742001. DOI: 10.1016/ j.partic.2016.08.003. URL: https://linkinghub.elsevier.com/retrieve/pii/ S167420011630133X (visited on 04/26/2023).
- [95] F. Usui and J.T. Carstensen. "Interactions in the Solid State I: Interactions of Sodium Bicarbonate and Tartaric Acid Under Compressed Conditions". In: *Journal of Pharmaceutical Sciences* 74.12 (Dec. 1985), pp. 1293–1297. ISSN: 00223549. DOI: 10.1002/jps.2600741209. URL: https: //linkinghub.elsevier.com/retrieve/pii/S0022354915469539 (visited on 05/09/2023).

- [96] Adam M. Stoklosa et al. "Effects of storage conditions, formulation, and particle size on moisture sorption and flowability of powders: A study of deliquescent ingredient blends". In: Food Research International 49.2 (Dec. 2012), pp. 783–791. ISSN: 09639969. DOI: 10.1016/j. foodres.2012.09.034. URL: https://linkinghub.elsevier.com/retrieve/pii/S09639969 12003936 (visited on 04/04/2023).
- [97] Rebecca A. Lipasek et al. "Effects of anticaking agents and storage conditions on the moisture sorption, caking, and flowability of deliquescent ingredients". In: *Food Research International* 45.1 (Jan. 2012), pp. 369–380. ISSN: 09639969. DOI: 10.1016/j.foodres.2011.10.037. URL: https://linkinghub.elsevier.com/retrieve/pii/S0963996911006181 (visited on 04/04/2023).
- [98] D. C. -H. Cheng. "The Effect of Pendular Moisture on the Tensile Strength of Powders". In: *The Journal of Adhesion* 2.2 (Apr. 1, 1970), pp. 82–92. ISSN: 0021-8464, 1545-5823. DOI: 10.1080/0021846708544583. URL: https://www.tandfonline.com/doi/full/10.1080/ 0021846708544583 (visited on 04/04/2023).
- [99] P York and N Pilpel. "The effect of temperature on the frictional, cohesive and electrical conducting properties of powders". In: *Materials Science and Engineering* 9 (Jan. 1972), pp. 281–291. ISSN: 00255416. DOI: 10.1016/0025-5416(72)90044-4. URL: https://linkinghub.elsevier.com/retrieve/pii/0025541672900444 (visited on 04/04/2023).
- [100] J R Britten and N Pilpel. "Effects of temperature on the tensile strength of pharmaceutical powders". In: Journal of Pharmacy and Pharmacology 30.1 (Apr. 12, 2011), pp. 673–677. ISSN: 2042-7158, 0022-3573. DOI: 10.1111/j.2042-7158.1978.tb13362.x. URL: https:// academic.oup.com/jpp/article/30/1/673/6197227 (visited on 04/04/2023).
- [101] C.L. Flemmer. "On the regime boundaries of moisture in granular materials". In: *Powder Technology* 66.2 (May 1991), pp. 191–194. ISSN: 00325910. DOI: 10.1016/0032-5910(91)80100-W. URL: https://linkinghub.elsevier.com/retrieve/pii/003259109180100W (visited on 04/04/2023).
- [102] N. Pilpel and J.R. Britten. "Effects of temperature on the flow and tensile strengths of powders". In: Powder Technology 22.1 (Jan. 1979), pp. 33–44. ISSN: 00325910. DOI: 10.1016/0032-591 0(79)85005-6. URL: https://linkinghub.elsevier.com/retrieve/pii/0032591079850056 (visited on 04/04/2023).
- [103] José Miguel Aguilera et al., eds. Food Engineering Interfaces. Food Engineering Series. New York, NY: Springer New York, 2011. ISBN: 978-1-4419-7474-7 978-1-4419-7475-4. DOI: 10. 1007/978-1-4419-7475-4. URL: http://link.springer.com/10.1007/978-1-4419-7475-4 (visited on 04/06/2023).
- [104] Yu Zou. "ANGLE OF INTERNAL FRICTION AND COHESION OF CONSOLIDATED GROUND MARIGOLD PETALS AS A FUNCTION OF PARTICLE SIZE, MOISTURE CONTENT, AND FLOW ENHANCER". PhD thesis. Guangzhou, China: South China University of Technology, 1993.
- [105] A.M. Hollenbach, M. Peleg, and R. Rufner. "Interparticle surface affinity and the bulk properties of conditioned powders". In: *Powder Technology* 35.1 (May 1983), pp. 51–62. ISSN: 00325910. DOI: 10.1016/0032-5910(83)85026-8. URL: https://linkinghub.elsevier.com/retrieve/ pii/0032591083850268 (visited on 04/06/2023).
- [106] Kailas K. Moravkar et al. "Assessment of pharmaceutical powders flowability and comparative evaluation of lubricants on development of gastro retentive tablets: An application of powder flow tester". In: Journal of Drug Delivery Science and Technology 71 (May 2022), p. 103265. ISSN: 17732247. DOI: 10.1016/j.jddst.2022.103265. URL: https://linkinghub.elsevier.com/ retrieve/pii/S1773224722001757 (visited on 04/06/2023).
- [107] Timothy C. Keener, George C. Frazier, and Wayne T. Davis. "THERMAL DECOMPOSITION OF SODIUM BICARBONATE". In: *Chemical Engineering Communications* 33.1 (Feb. 1985), pp. 93–105. ISSN: 0098-6445, 1563-5201. DOI: 10.1080/00986448508911162. URL: https: //www.tandfonline.com/doi/full/10.1080/00986448508911162 (visited on 04/06/2023).

- [108] L. N. Gard and R. C. Butler. "Determination of Moisture in Sodium Bicarbonate". In: Analytical Chemistry 26.8 (Aug. 16, 1954), pp. 1367–1368. ISSN: 0003-2700, 1520-6882. DOI: 10.1021/ ac60092a033. URL: https://pubs.acs.org/doi/abs/10.1021/ac60092a033 (visited on 04/06/2023).
- [109] W Kuu. "Effect of relative humidity and temperature on moisture sorption and stability of sodium bicarbonate powder". In: *International Journal of Pharmaceutics* 166.2 (May 18, 1998), pp. 167–175. ISSN: 03785173. DOI: 10.1016/S0378-5173(98)00049-0. URL: https://linkinghub.elsevier.com/retrieve/pii/S0378517398000490 (visited on 04/06/2023).
- [110] Hong Lee Lim, Karen P. Hapgood, and Brian Haig. "Understanding and preventing agglomeration in a filter drying process". In: *Powder Technology* 300 (Oct. 2016), pp. 146–156. ISSN: 00325910. DOI: 10.1016/j.powtec.2016.03.003. URL: https://linkinghub.elsevier. com/retrieve/pii/S0032591016300961 (visited on 04/06/2023).
- [111] E Teunou, J J Fitzpatrick, and E C Synnott. "Characterisation of food powder owability". In: *Journal of Food Engineering* (1999).
- [112] V. Ganesan, K.A. Rosentrater, and K. Muthukumarappan. "Flowability and handling characteristics of bulk solids and powders – a review with implications for DDGS". In: *Biosystems Engineering* 101.4 (Dec. 2008), pp. 425–435. ISSN: 15375110. DOI: 10.1016/j.biosystemseng. 2008.09.008. URL: https://linkinghub.elsevier.com/retrieve/pii/S1537511008002791 (visited on 04/26/2023).
- S.B. Tan and J.M. Newton. "Powder flowability as an indication of capsule filling performance". In: International Journal of Pharmaceutics 61.1 (June 1990), pp. 145–155. ISSN: 03785173. DOI: 10.1016/0378-5173(90)90053-7. URL: https://linkinghub.elsevier.com/retrieve/ pii/0378517390900537 (visited on 04/26/2023).
- [114] Imran M. Zainuddin et al. "Experimental Study on Powder Flowability Using Vibration Shear Tube Method". In: Particle & Particle Systems Characterization 29.1 (May 2012), pp. 8–15. ISSN: 09340866. DOI: 10.1002/ppsc.201100052. URL: https://onlinelibrary.wiley.com/doi/ 10.1002/ppsc.201100052 (visited on 04/26/2023).
- [115] Fabio Ippolito et al. "Impact of Bimodal Particle Size Distribution Ratio of Functional Calcium Carbonate Filler on Thermal and Flowability Properties of Polyamide 12". In: Applied Sciences 11.2 (Jan. 11, 2021), p. 641. ISSN: 2076-3417. DOI: 10.3390/app11020641. URL: https:// www.mdpi.com/2076-3417/11/2/641 (visited on 04/26/2023).
- [116] L. J. Johnston, J. Goihl, and G. C. Shurson. "Selected Additives Did Not Improve Flowability of DDGS in Commercial Systems". In: *Applied Engineering in Agriculture* 25.1 (2009), pp. 75–82. ISSN: 1943-7838. DOI: 10.13031/2013.25422. URL: http://elibrary.asabe.org/abstract. asp??JID=3&AID=25422&CID=aeaj2009&v=25&i=1&T=1 (visited on 04/26/2023).
- [117] Irene Luna-Guzmán and Diane M Barrett. "Comparison of calcium chloride and calcium lactate effectiveness in maintaining shelf stability and quality of fresh-cut cantaloupes". In: *Postharvest Biology and Technology* 19.1 (May 2000), pp. 61–72. ISSN: 09255214. DOI: 10.1016/S0925-5214(00)00079-X. URL: https://linkinghub.elsevier.com/retrieve/pii/S092552140000 079X (visited on 05/09/2023).
- [118] G. Fraissler et al. "Influence of dry and humid gaseous atmosphere on the thermal decomposition of calcium chloride and its impact on the remove of heavy metals by chlorination". In: *Chemical Engineering and Processing: Process Intensification* 48.1 (Jan. 2009), pp. 380–388. ISSN: 02552701. DOI: 10.1016/j.cep.2008.05.003. URL: https://linkinghub.elsevier. com/retrieve/pii/S0255270108001153 (visited on 05/09/2023).
- [119] Zhao-Fen Jin et al. "Thermal and water storage characteristics of super-absorbent polymer gel which absorbed aqueous solution of calcium chloride". In: International Journal of Heat and Mass Transfer 43.18 (Sept. 2000), pp. 3407–3415. ISSN: 00179310. DOI: 10.1016/S0017-9310(99)00372-5. URL: https://linkinghub.elsevier.com/retrieve/pii/S001793109900 3725 (visited on 05/09/2023).

- [120] M. Gaeini et al. "Characterization of microencapsulated and impregnated porous host materials based on calcium chloride for thermochemical energy storage". In: Applied Energy 212 (Feb. 2018), pp. 1165–1177. ISSN: 03062619. DOI: 10.1016/j.apenergy.2017.12.131. URL: https: //linkinghub.elsevier.com/retrieve/pii/S0306261917318548 (visited on 05/09/2023).
- [121] J Food Process Engineering 2022 Juarez Enriquez A review on the influence of water on food powder flowability.pdf.
- [122] S.B. Tan and J.M. Newton. "Influence of capsule dosator wall texture and powder properties on the angle of wall friction and powder-wall adhesion". In: *International Journal of Pharmaceutics* 64.2 (Oct. 1990), pp. 227–234. ISSN: 03785173. DOI: 10.1016/0378-5173(90)90273-7. URL: https://linkinghub.elsevier.com/retrieve/pii/0378517390902737 (visited on 05/09/2023).
- [123] Hrishikesh Mitra et al. "Influence of moisture content on the flow properties of basundi mix". In: *Powder Technology* 312 (May 2017), pp. 133–143. ISSN: 00325910. DOI: 10.1016/j.powtec. 2017.02.039. URL: https://linkinghub.elsevier.com/retrieve/pii/S0032591017301675 (visited on 05/09/2023).
- [124] Erica Emery et al. "Flowability of moist pharmaceutical powders". In: Powder Technology 189.3 (Feb. 2009), pp. 409–415. ISSN: 00325910. DOI: 10.1016/j.powtec.2008.06.017. URL: htt ps://linkinghub.elsevier.com/retrieve/pii/S0032591008003562 (visited on 05/09/2023).
- [125] C. Larrigaudière et al. "Storage performance of clementines treated with hot water, sodium carbonate and sodium bicarbonate dips". In: *The Journal of Horticultural Science and Biotechnology* 77.3 (Jan. 2002), pp. 314–319. ISSN: 1462-0316, 2380-4084. DOI: 10.1080/14620316. 2002.11511499. URL: http://www.tandfonline.com/doi/full/10.1080/14620316.2002.11511499 (visited on 05/09/2023).
- [126] S. C. S. Rocha and J. F. Nunes. "Agglomeration of Sodium Bicarbonate in Vibrofluidized Bed: Process Analysis and Evaluation of CO ₂ Release". In: *Drying Technology* 30.11 (Sept. 2012), pp. 1332–1341. ISSN: 0737-3937, 1532-2300. DOI: 10.1080/07373937.2012.689403. URL: http://www.tandfonline.com/doi/abs/10.1080/07373937.2012.689403 (visited on 05/09/2023).
- [127] C.E. Capes. "The correlation of agglomerate strength with size". In: *Powder Technology* 5.2 (Jan. 1972), pp. 119–125. ISSN: 00325910. DOI: 10.1016/0032-5910(72)80011-1. URL: https://linkinghub.elsevier.com/retrieve/pii/0032591072800111 (visited on 05/09/2023).
- [128] Peter Hamilton et al. "Studies of particle drying using non-invasive Raman spectrometry and particle size analysis". In: *The Analyst* 136.10 (2011), p. 2168. ISSN: 0003-2654, 1364-5528. DOI: 10.1039/c0an00893a. URL: http://xlink.rsc.org/?DOI=c0an00893a (visited on 05/09/2023).
- [129] Wolfgang B. Pietsch. "The strength of agglomerates bound by salt bridges". In: *The Canadian Journal of Chemical Engineering* 47.4 (Aug. 1969), pp. 403–409. ISSN: 00084034, 1939019X. DOI: 10.1002/cjce.5450470419. URL: https://onlinelibrary.wiley.com/doi/10.1002/cjce.5450470419 (visited on 05/09/2023).
- [130] Z. Pan, S. Zhang, and J. Jane. "Effects of Extrusion Variables and Chemicals on the Properties of Starch-Based Binders and Processing Conditions". In: *Cereal Chemistry Journal* 75.4 (July 1998), pp. 541–546. ISSN: 0009-0352. DOI: 10.1094/CCHEM.1998.75.4.541. URL: http: //doi.wiley.com/10.1094/CCHEM.1998.75.4.541 (visited on 05/09/2023).
- [131] G.H. Rye, P.E. Neumann, and C.E. Walker. *Effects of some baking ingredients on physical and structural properties of wheat flour extrudates.*
- [132] John E. Sani, Roland Kufre Etim, and Alexander Joseph. "Compaction Behaviour of Lateritic Soil–Calcium Chloride Mixtures". In: *Geotechnical and Geological Engineering* 37.4 (Aug. 2019), pp. 2343–2362. ISSN: 0960-3182, 1573-1529. DOI: 10.1007/s10706-018-00760-6. URL: http://link.springer.com/10.1007/s10706-018-00760-6 (visited on 05/09/2023).

- [133] Christer Nyström et al. "Bonding Surface area and Bonding Mechanism-Two Important Factors fir the Understanding of Powder Comparability". In: *Drug Development and Industrial Pharmacy* 19.17 (Jan. 1993), pp. 2143–2196. ISSN: 0363-9045, 1520-5762. DOI: 10.3109/036390493090 47189. URL: http://www.tandfonline.com/doi/full/10.3109/03639049309047189 (visited on 05/10/2023).
- [134] François Lavoie, Louis Cartilier, and Roch Thibert. "New Methods Characterizing Avalanche Behavior to Determine Powder Flow". In: ().
- [135] Volker Kehlenbeck. *Continuous dynamic mixing of cohesive powders*. Berichte aus der Verfahrenstechnik. Aachen: Shaker, 2007. 252 pp. ISBN: 978-3-8322-5726-2.
- [136] Gerard R. Klinzing and Antonios Zavaliangos. "A Simplified Model of Moisture Transport in Hydrophilic Porous Media With Applications to Pharmaceutical Tablets". In: *Journal of Pharmaceutical Sciences* 105.8 (Aug. 2016), pp. 2410–2418. ISSN: 00223549. DOI: 10.1016/j.xphs. 2016.05.030. URL: https://linkinghub.elsevier.com/retrieve/pii/S0022354916414668 (visited on 05/10/2023).
- [137] T Miller and P York. "Physical and chemical characteristics of some high purity magnesium stearate and palmitate powders". In: *International Journal of Pharmaceutics* 23.1 (Jan. 1985), pp. 55–67. ISSN: 03785173. DOI: 10.1016/0378-5173(85)90222-4. URL: https://linkingh ub.elsevier.com/retrieve/pii/0378517385902224 (visited on 05/10/2023).
- [138] Rumela Bhadra, K Muthukumarappan, and Kurt A Rosentrater. "Effects of Varying CDS levels and Drying and Cooling Temperatures on Flowability Properties of DDGS". In: ().
- [139] A.P. Tinke et al. "A review of underlying fundamentals in a wet dispersion size analysis of powders". In: *Powder Technology* 196.2 (Dec. 2009), pp. 102–114. ISSN: 00325910. DOI: 10. 1016/j.powtec.2009.08.005. URL: https://linkinghub.elsevier.com/retrieve/pii/ S0032591009004628 (visited on 05/10/2023).
- [140] Hao Shi, Stefan Luding, and Vanessa Magnanimo. "Limestone Powders Yielding and Steady State Resistance under Shearing with Different Testers". In: ().
- [141] Anthony H.J. Paterson and John E. Bronlund. "The practical implications of temperature induced moisture migration in bulk lactose". In: *Journal of Food Engineering* 91.1 (Mar. 2009), pp. 85–90. ISSN: 02608774. DOI: 10.1016/j.jfoodeng.2008.08.021. URL: https://linkinghub.elsevier.com/retrieve/pii/S0260877408003956 (visited on 05/10/2023).
- [142] Ertan Ermiş, ed. Food Powders Properties and Characterization. Food Engineering Series. Cham: Springer International Publishing, 2021. ISBN: 978-3-030-48907-6 978-3-030-48908-3. DOI: 10.1007/978-3-030-48908-3. URL: https://link.springer.com/10.1007/978-3-030-48908-3 (visited on 05/10/2023).
- [143] Jamshiya.E* {And} Anju.P. "Formulation, Evaluation And Optimization Of Floating Matrix Tablets Of Carvedilol". In: (Mar. 5, 2018). Publisher: Zenodo. DOI: 10.5281/ZENODO.1188894. URL: https://zenodo.org/record/1188894 (visited on 05/10/2023).
- [144] Hao Hou and Changquan Calvin Sun. "Quantifying Effects of Particulate Properties on Powder Flow Properties Using a Ring Shear Tester". In: *Journal of Pharmaceutical Sciences* 97.9 (Sept. 2008), pp. 4030–4039. ISSN: 00223549. DOI: 10.1002/jps.21288. URL: https://linkinghub. elsevier.com/retrieve/pii/S0022354916326880 (visited on 05/10/2023).
- [145] Hao Shi et al. "Effect of Particle Size and Cohesion on Powder Yielding and Flow". In: KONA Powder and Particle Journal 35.0 (2018), pp. 226–250. ISSN: 0288-4534, 2187-5537. DOI: 10.14356/kona.2018014. URL: https://www.jstage.jst.go.jp/article/kona/35/0/35_ 2018014/_article (visited on 05/10/2023).
- [146] Sharad Mangal, B Pharm, and M Pharm. "Co-processing Pharmaceutical Ingredients to Engineer Novel Multifunctional Excipients". In: ().
- [147] Christina Gustafsson. Solid state characterisation and compactation behaviour of pharmaceutical materials. Comprehensive Summaries of Uppsala Dissertations from the Faculty of Pharmacy 233. Uppsala: AUU, 2000. 64 pp. ISBN: 978-91-554-4813-4.
- [148] James Hallie Cain. Characterization of Bulk Material Behavior for Granular Materials.

- [149] P. Gopalkrishnan, I. Manas-Zloczower, and D.L. Feke. "Effect of morphology and extent of infiltration on the cohesivity and dispersion mechanisms of particle agglomerates". In: *Chemical Engineering Science* 62.14 (July 2007), pp. 3740–3747. ISSN: 00092509. DOI: 10.1016/j.ces. 2007.03.037. URL: https://linkinghub.elsevier.com/retrieve/pii/S0009250907003065 (visited on 05/10/2023).
- [150] P. Levresse, I. Manas-Zloczower, and D. L. Feke. "Dispersion Studies of Agglomerates in Steady and Dynamic Flows of Polymeric Materials". In: *Rubber Chemistry and Technology* 75.1 (Mar. 1, 2002), pp. 119–132. ISSN: 1943-4804, 0035-9475. DOI: 10.5254/1.3547664. URL: https: //meridian.allenpress.com/rct/article/75/1/119/92972/Dispersion-Studies-of-Agglomerates-in-Steady-and (visited on 05/10/2023).
- [151] Christof Krüger. "Sugar and bulk sweeteners". In: Beckett's Industrial Chocolate Manufacture and Use. Ed. by Stephen T. Beckett, Mark S. Fowler, and Gregory R. Ziegler. Chichester, UK: John Wiley & Sons, Ltd, Mar. 7, 2017, pp. 72–101. ISBN: 978-1-118-92359-7 978-1-118-78014-5. DOI: 10.1002/9781118923597.ch4. URL: https://onlinelibrary.wiley.com/doi/10.1002/9781118923597.ch4 (visited on 05/10/2023).
- [152] M. Bumiller, J.J. Carson, and Prescott. A preliminary investigation concerning the effect of particle shape on a powder's flow properties. URL: https://www.researchgate.net/publication/ 267938447_A_PRELIMINARY_INVESTIGATION_CONCERNING_THE_EFFECT_OF_PARTICLE_SHAPE_ ON_A_POWDER%27S_FLOW_PROPERTIES (visited on 05/29/2023).
- [153] TULib. Searching and Resources. 2023. URL: https://tulib.tudelft.nl/searching-resou rces/ (visited on 06/16/2023).

A

Additional Information

A.1. Search Strategy

An abundance of papers and book excerpts have been analysed in an attempt to gather information on the effect of operational conditions on the mechanical characteristics of various materials. On top of the sources named in Table 5.1, the following papers have been analysed in depth. It was initially expected that these papers would be helpful for the completion of this research, however, these papers did not provide any helpful insights into the subject. An analysis of the search strategy used throughout this report is provided below, to illustrate the effort that has been put in finding relevant source material.

Researched sources: [96] [97] [98] [99] [100] [101] [102] [103] [104] [105] [106] [107] [108] [109] [110] [111] [112] [113] [114] [115] [116] [117] [118] [119] [120] [121] [122] [123] [124] [125] [126] [127] [128] [129] [130] [131] [132] [133] [134] [135] [136] [137] [138] [139] [140] [141] [142] [143] [144] [145] [146] [147] [148] [149] [150] [151] [152]

Throughout this literature research, the helpful tips available on the TU Delft library website have been used to structure the searching process [153]. A search plan was created and a list of important terms and their synonyms was constructed. Search queries that were used throughout this research looked as follows:



Figure A.1: Search queries.



An example of this is given below:

Figure A.2: Example of used search queries.